Author's response to reviews

Title: The disease burden of pertussis in adults 50 years old and older in the United States: a retrospective study

Authors:

Catherine Balderston McGuiness (kmcguiness@us.imshealth.com)
Jerrold W Hill (j.hill@us.imshealth.com)
Eileen M Fonseca (efonseca@us.imshealth.com)
Gregory P Hess (ghess@us.imshealth.com)
William P Hitchcock (lajollapeds@yahoo.com)
Girishanthy X Krishnarajah (girishanthy.x.krishnarajah@gsk.com)

Version: 3 Date: 30 September 2012

Author's response to reviews: see over
September 30, 2012

Editorial Office
BMC Infectious Diseases

Dear Joanne Appleford-Cook:

Please find attached resubmission of manuscript number 753161337092448, "The disease burden of pertussis in adults ≥50 years old in the United States: a retrospective study" for consideration in BMC Infectious Diseases.

We are pleased by the enthusiasm shown for this project, and for the positive reception of our work. We are grateful for the opportunity to revise our paper and thank the reviewers for their helpful observations, comments, and suggestions for improvement. In the attached document, we have detailed our responses to the reviewers’ feedback.

We believe that we have been able to address the reviewer’s comments and are convinced that our manuscript is now suitable for publication in BMC Infectious Diseases.

Please direct correspondence concerning this manuscript to Catherine Balderston McGuiness: IMS Health, One IMS Drive, Plymouth Meeting, PA, USA; e-mail: kmcguiness@us.imshealth.com; telephone +001 610-834-0800; or fax +001 610-834-8817.

Yours Sincerely,

Catherine Balderston McGuiness